Author's response to reviews

Title: Priority setting in primary health care - dilemmas and opportunities: a focus group study

Authors:

Eva Arvidsson (eva.arvidsson@ltkalmar.se)
Malin André (malin.andre@ltdalarna.se)
Lars Borgquist (lars.borgquist@liu.se)
Per Carlsson (per.carlsson@liu.se)

Version: 4 Date: 5 August 2010

Author's response to reviews: see over
We are delighted to hear that you are prepared to publish our paper “Priority Setting in Primary Health Care – Dilemmas and Opportunities: A Focus Group Study” (MS: 1245552833375119). We hereby submit a new version of our paper which include the requested formatting revisions.

Author list has been reversed.
Title has been reversed.
Cities have been included in the affiliation details.
All text is in black and all highlighting has been removed.
Authors’ contribution has been reversed.
Reference 32 has been reversed.
Vertical lines have been removed from the tables and Legend text have been placed below the tables.

Yours sincerely,

Eva Arvidsson